Author’s response to reviews

Title: Primary central nervous system small lymphocytic lymphoma in the bilateral ventricles: two case reports

Authors:

Rongjing Guo (guorongjing533@126.com)
Xiaolong Zhang (long7878521@163.com)
Chunxiao Niu (409433694@qq.com)
Yibin Xi (xiyibin@fmmu.edu.cn)
Hong Yin (yinhong@fmmu.edu.cn)
Hong Lin (tdneuroa@fmmu.edu.cn)
ting chang (changting1981@163.com)

Version: 1 Date: 27 Apr 2019

Author’s response to reviews:

Dear editor Anne,

Thank you for your advise. We have revised our manuscript according to your request. In the revised manuscript, we do not use the term "BenZ". I would like to express my great appreciation to your comments and looking forward to hearing from you.

Thank you and best regards,

Ting Chang